Author's response to reviews

Title: The Assessment of Recalled Parental Rearing and its Relationship to Life Satisfaction and Interpersonal Problems: A General Population Study

Authors:

Katja Petrowski (katja.petrowski@tu-dresden.de)
Henrik Berth (berth@inhaltsanalyse.de)
Silke Schmidt (sischmid@uke.uni-hamburg.de)
Joerg Schumacher (joerg.schumacher@email.de)
Andreas Hinz (hina@medizin.uni-leipzig.de)
Elmar Braehler (elmar.braehler@medizin.uni-leipzig.de)

Version: 4 Date: 18 November 2008

Author's response to reviews: see over
Resubmission of revised article

Dear BioMed Central Editorial Team,

attached please find the article (Ms-No: 8421314951667679).

As you recommended in the last mail we added a detailed statement concerning the ethical approval in the Method section. An Acknowledgement section for our article is not necessary since all contributing persons are listed as authors of the article. Furthermore, we revised the article concerning your journal style requirements.

I confirm that this manuscript has not been previously published and has not been nor will be submitted elsewhere during the review process.

Sincerely,

Katja Petrowski